Author's response to reviews

Title: Alpha-Glucosidase Inhibitory Activity and Antioxidant Activity of Flavonoids Compound and Triterpenoid Compound from Agrimonia Pilosa Ledeb.

Authors:

Xi Liu (liuxi900626@126.com)
Liancai Zhu (zhuliancai75@126.com)
Jun Tan (tanjunmail@126.com)
Xuemei Zhou (zhouxuemei15206@yahoo.cn)
Bochu Wang (wangbc2000@126.com)
Xian Yang (yyxdsp@163.com)

Version: 4
Date: 17 December 2013

Author's response to reviews: see over
Dear Editor,

We are highly appreciated for these helpful reviews from the reviewers about our manuscript entitled “α-Glucosidase Inhibitory Activity and Antioxidant Activity of Flavonoid Compound and Triterpenoid Compound from Agrimonia Pilosa Ledeb.” (MS: 4023323951046452). We have revised carefully for whole manuscript according to reviewers’ comments. The major revised portions and response to reviewers’ comments were listed below. We hope that the revised manuscript is acceptable for publication.

Thank you very much for your continued attention.

Best wishes,

Liancai Zhu

Response to reviewers’ comments:

Reviewer 1

1. The statics is essential to support your conclusion.

**Answer:** Thanks for your constructive suggestion. The statistics have been done and analyzed by using SPSS 19.0 software to support our conclusion. Please see the “Statistical Analysis” in the “Methods” section (page 12 and 13) and Table 3.

2. The presentation is bad.

**Answer:** The presentation has been improved by Min Luo who is the Associate Editor of Journal of Chongqing University-English Edition.

Reviewer 2

Page 5, the second paragraph, “Agrimonia pilosa Ledeb” should be corrected as “Agrimonia pilosa Ledeb”.

**Answer:** Thanks for your suggestion. “Agrimonia pilosa Ledeb” has been corrected as “Agrimonia pilosa Ledeb” in the full text.

Reviewer 3

My decision is to accept the corrected manuscript for to be published by BMC Complementary and alternative medicine because the authors have responded satisfactorily giving explanations and making the necessary changes to remove the objections expressed in the original manuscript.

**Answer:** Thanks for your comment.